Author's response to reviews

Title: Determination of iron sucrose (Venofer) or iron dextran (DexFerrum) removal by hemodialysis: an an-vitro study

Authors:

Dr Harold J Manley (manleyh@umkc.edu)
Dr Darren W Grabe (graped@acp.edu)

Version: 4 Date: 9 Jan 2004

The formatting changes requested (listed below) were implemented.
-------------------------------------
Title - Please remove all unnecessary capitalisation from the title.
Headers - Please remove all headers.
Qualifications - Please remove the authors' qualifications (e.g. PhD) from the title page.
References - References should be cited in the text using consecutive numbers in square brackets, inline and placed before the full stop.
Section headings - Please remove all unnecessary capitalisation.